Author’s response to reviews

**Title:** Clinical Characteristics and Outcomes of the Oldest Old people with Type 2 Diabetes - Perspective from a tertiary diabetes center in Thailand

**Authors:**

YOTSAPON THEWJITCHAROEN (kamijoa@hotmail.com)

SIRINATE KRITTIYAWONG (sirinate_k@yahoo.com)

EKGALUCK WANOTHAYAROJ (ekgaluck@yahoo.com)

SOMBOON VONGTERAPAK (vongterapak@hotmail.com)

TAWEE ANUNTAKULNATEE (taweeanunt@yahoo.com)

WORAWIT KITTIPOOOM (wkittipoom@yahoo.com)

SOONTAREE NAKASATIEN (soontareenaka@gmail.com)

THEP HIMATHONGKAM (thepjudy@loxinfo.co.th)

**Version:** 2 **Date:** 24 May 2016

**Author’s response to reviews:**

We edited a declarations section as suggested from editorial.